Author's response to reviews

Title: Pluripotency-associated genes in human nasopharyngeal carcinoma CNE-2 cells are reactivated by a unique epigenetic sub-microenvironment

Authors:

Jun-Xia Cao (jun-xia.cao@jax.org)
Yu-Xin Cui (yx cui2000@yahoo.com)
Zi-Jie Long (bio_longzijie@yahoo.com.cn)
Zhong-Min Dai (zhongmindai@hotmail.com)
Ji-Yan Lin (doctorlinjiyan@yahoo.com.cn)
Yi Liang (liangyi_best@163.com)
Fei-Meng Zheng (mongolia8206@yahoo.com.cn)
Yi-Xin Zeng (yxzeng@gzsums.edu.cn)
Quentin Liu (liuq9@mail.sysu.edu.cn)

Version: 5 Date: 22 February 2010

Author's response to reviews: see over
Dear Professor,

We would like to give our sincere appreciation to you for your valuable time and effort reviewing our work. Now we submit our revised manuscript in your prestigious Journal. We declare that we have no competing interests.

As suggested by you, we have made the following revision point by point:

1. Figures: All figures must be cited; figures not cited will be excluded from the final manuscript. Currently we can not find figure 4 cited. Answer: Sorry. We have cited figure 4 in result section. Thank you very much.

2. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. Answer: We have carefully checked the whole manuscript again. Thanks very much.

Again, we appreciate your time and effort for examining our work.
Sincerely yours,

Quentin Liu, MD, PhD

Prof. of Cancer Biology